Author's response to reviews

Title: Modernising the regulation of medical migration: moving from national monopolies to international markets

Authors:

Richard J Epstein (repstein@stvincents.com.au)
Stephen D Epstein (epstein@nigelbowen.com.au)

Version: 7 Date: 6 August 2012

Author's response to reviews: see over
Authors' responses to Michelle McLean's comments

Wherever deemed possible by the authors, the changes of phrasing ("language needs to be addressed") suggested by the reviewer have been made (p. 1-13).

The citation within the text for Figure 1 has been brought forward as suggested to the beginning of the Discussion (p. 5).

The citation within the text for Table 1 has been brought forward as suggested, to the end of the Discussion (p. 10).

More detail has been inserted into the Legend for Figure 1, as suggested (p. 13).

The Journal publishes Debates; it does not publish Commentaries (see BMC Instructions for Authors).